Author's response to reviews

Title: Anxiety and stress in the postpartum: Is there more to postnatal distress than depression?

Authors:

   Renee L Miller (r.miller@bigpond.net.au)
   Julie F Pallant (jpallant@swin.edu.au)
   Lisa M Negri (lisa.m.negri@rmit.edu.au)

Version: 3 Date: 4 March 2006

Author's response to reviews: see over
Anxiety and stress in the postpartum: Is there more to postnatal distress than depression?
Renee L Miller, Julie F Pallant and Lisa M Negri

Title & Abstract – We have made sure that these are identical in the manuscript and in the submission system.

Author list - We have made sure that these are identical in the manuscript and in the submission system.

Affiliations – Full details of affiliations are provided. Please note that RMIT University is the full official title of the affiliation of the third author.

Section headings - All headings and sub-headings in the manuscript have been changed to sentence case with all unnecessary capitals removed.

Author’s contributions – This has been changed to Authors' contributions.

Figure title - The image files have been changed to remove the title.

Additional minor changes were made in response to the reviewers’ comments received on the revised manuscript.